Author's response to reviews

Title: Drug-resistance in Streptococcus pneumoniae isolates among Spanish middle aged and older adults with community-acquired pneumonia.

Authors:
Angel Vila-Corcoles (avila.tarte.ics@gencat.net)
Ferran Bejarano-Romero (fbejarano.tarte.ics@gencat.cat)
Elisabeth Salsench (eli.ss@terra.es)
Olga Ochoa-Gondar (oochoa.tarte.ics@gencat.cat)
Cinta de Diego (cintaddc@tinet.org)
Frederic Gomez-Bertomeu (ffgomez.hj23.ics@gencat.cat)
Xavier Raga-Luria (xraga@xarxatecla.cat)
Xavier Cliville-Guasch (xcliville@xarxatecla.cat)
Victoria Arija (victoria.arija@urv.cat)

Version: 5 Date: 20 March 2009

Author's response to reviews: see over
Dear editor,

After revision of the accepted version of the manuscript entitled Drug-resistance in Streptococcus pneumoniae isolates among Spanish middle aged and older adults with community-acquired pneumonia (ID 1408777051232562) we have corrected two orthographic mistakes:

1. Farmacology Unit (in the title page) was changed by Pharmacology Unit.
2. Contemporany isolates (in the table 3) was changed by Contemporary isolates.

Yours sincerely,

Angel Vila Corcoles